Author's response to reviews

Title: Vitamin C as an adjuvant for treating major depressive disorder and suicidal behavior, a randomized placebo controlled clinical trial

Authors:

Ali sahraian (shraian@sums.ac.ir)
Ahmad Ghanziadeh (ghanizadeha@hotmail.com)
Feresteh Kazemeini (kazemini@sums.ac.ie)

Version: 5 Date: 3 February 2015

Author's response to reviews: see over
Dear Editors-in-Chief: Doug Altman, Curt Furberg, Jeremy Grimshaw

MS: 169986155136750
Vitamin C as an adjuvant for treating major depressive disorder and suicidal behavior, a randomized placebo controlled clinical trial
Ali sahraian, Ahmad Ghanziadeh and Feresteh Kazemeini

Thank you for your comments.

Comment:
"Please correct some type errors. eg.: In the abstract and result section, a sentence start with a number which should be spelled out, there are spaces between words missing and a paragraph seems to be underlined whereas this should not be the case (p.5 and p. 10)."
Response: Those are corrected.

Comment: Editorial request:
1. Please include a figure title and legend section after the reference list.

Response: It is added.

The Best
Ahmad Ghanizadeh